

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Euwals, Rob; Ward, Melanie E.

Working Paper The Remuneration of British Academics

IZA Discussion Papers, No. 178

Provided in Cooperation with: IZA – Institute of Labor Economics

Suggested Citation: Euwals, Rob; Ward, Melanie E. (2000) : The Remuneration of British Academics, IZA Discussion Papers, No. 178, Institute for the Study of Labor (IZA), Bonn

This Version is available at: https://hdl.handle.net/10419/21021

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

IZA DP No. 178

The Remuneration of British Academics

Rob Euwals Melanie Ward

July 2000

Forschungsinstitut zur Zukunft der Arbeit Institute for the Study of Labor

The Renumeration of British Academics

Rob Euwals

IZA, Bonn

Melanie Ward

IZA, Bonn

Discussion Paper No. 178 July 2000

IZA

P.O. Box 7240 D-53072 Bonn Germany

Tel.: +49-228-3894-0 Fax: +49-228-3894-210 Email: iza@iza.org

This Discussion Paper is issued within the framework of IZA's research area *The Future of Work.* Any opinions expressed here are those of the author(s) and not those of the institute. Research disseminated by IZA may include views on policy, but the institute itself takes no institutional policy positions.

The Institute for the Study of Labor (IZA) in Bonn is a local and virtual international research center and a place of communication between science, politics and business. IZA is an independent, nonprofit limited liability company (Gesellschaft mit beschränkter Haftung) supported by the Deutsche Post AG. The center is associated with the University of Bonn and offers a stimulating research environment through its research networks, research support, and visitors and doctoral programs. IZA engages in (i) original and internationally competitive research in all fields of labor economics, (ii) development of policy concepts, and (iii) dissemination of research results and concepts to the interested public. The current research program deals with (1) mobility and flexibility of labor markets, (2) internationalization of labor markets in transition, (5) the future of work, (6) project evaluation and (7) general labor economics.

IZA Discussion Papers often represent preliminary work and are circulated to encourage discussion. Citation of such a paper should account for its provisional character.

IZA Discussion Paper No. 178 July 2000

ABSTRACT

The Renumeration of British Academics^{*}

This paper examines both pay relativities and mechanisms for pay determination within the UK academic labour market drawing upon a particularly detailed data set of 635 academics from five traditional Scottish Universities. In the existing literature, the fact that in many occupations, employees are paid according to explicitly determined wage scales is mostly ignored. We employ salary, grade and spinal point information to incorporate the fixed framework of academic salaries into analysis. Our results outline the importance of individual productivity, measured through publication, grant receipt and teaching skill, in attracting financial reward. We find a large penalty associated with time out of the profession and evidence for the deregulation of established pay and promotion structures. In order to identify those academics most likely to leave the profession, analysis also considers the determinants of individuals' reservation and deserved salary. Controlling for individual characteristics we find that lecturers hold the lowest reservation salaries in relation to their current salary level. The academic profession is therefore most at risk from loosing its staff at this grade. We find however no (self-)selection on the basis of the productivity of individuals.

JEL Classification: A22, C35, J31, J44

Keywords: Academic labour market, salary, salary scales

Rob Euwals IZA P.O. Box 7240 53072 Bonn Germany Tel.: +49-228-3894-302 Fax: +49-228-3894-210 Email: euwals@iza.org

^{*} We wish to thank our colleagues at IZA, and the seminar-participants at Leicester University, Essex University and ESPE 2000, Bonn, for valuable comments and helpful discussions.

1. Introduction

The Hay report (1997) revealed that over the previous 10 years, university pay dropped by as much as 20% in real terms, falling behind comparable professions in the public sector. Dearing (1997) confirmed this trend, advising that although academic remuneration should be sufficient to recruit, retain and motivate staff of the required quality, the majority of staff in higher education were in fact paid substantially below comparable private and public sector rates. Although one might argue that academics are interested in more than mere pecuniary reward,¹ long-term underpayment may spark the drain of high quality individuals from academia into more lucrative positions in the private sector, or to academic positions abroad. Strikes over pay during 1996 and 1999 were suggestive of a profession on the edge of their salary threshold.

Within the context of the academic underpayment debate, the adequacy of the remuneration structure currently in place has been held to question. Establishment level academic salaries, at least below professorial grades, remain formally set through a nationally negotiated fixed salary structure agreed between the Association of University Teachers (AUT) and the University and College's Employers Association (UCEA). Within this formal framework, staff progression is largely automatic and dependent on years of work. With the emergence of the research and teaching assessment exercises and the dramatic increase in student numbers over recent years, the importance of publication and administrative responsibility within the academic job has been consolidated. It is unclear, however, whether the current reward system adequately recognises individual productivity.

This paper examines both pay relativities and mechanisms for pay determination within the UK academic profession and aims to provide some visibility into academic reward. We utilise a unique cross sectional dataset, which includes detailed information on salary and grade of the academic staff of five old established universities. An important feature of the data is that it includes measures for individual research productivity. Such detailed data is scarce in the existing literature in this area. In the first step of our analysis, we employ salary, grade and spinal point information to incorporate the fixed framework of salary scales into analysis. Although

¹ Academics have after all undertaken periods of extended study relative to the general labour force at an opportunity cost of perhaps considerable foregone earnings. For further discussion of non pecuniary reward see Ward and Sloane (1999).

there exists an extensive literature on pay schemes, the fact that in many occupations employees are paid according to explicitly determined wage scales is mostly ignored. Our framework allows us to consider the effect of policy-changes to the current system of academic reward, such as changing the wage rates within the fixed framework, or the effect of changing the fixed framework itself. In the second step of our analysis, we investigate the determinants of academics' deserved and reservation salary with the aim of identifying the most mobile and/or most dissatisfied staff within the profession. Although the mobility of staff in and out of the academic sector might be beneficial to some extent, the profession should ensure its capability to retain the best and most productive individuals. We will try to identify those who are most at risk from being lost to the profession.

The remainder of the paper is structured as follows: Section 2 summarizes previous literature relevant to our analysis and section 3 outlines the main characteristics of the dataset used in this paper. Section 4 introduces our model, which incorporates the fixed framework of salary scales, for the analysis of the determinants of actual academic salary. Section 5 considers a model of deserved and reservation salary. Section 6 concludes.

2. Previous literature

Until recently there had been virtually no work written on pay within the British academic profession. The lack of detailed data on academics in the UK has provided a hurdle for potential researchers. National statistics, collected by the Universities Statistical Record and the Higher Educational Statistics Agency, contain only very limited information. The census of academic salaries collected data on gender, age, date of recruitment, rank, faculty and salary, but ceased in 1993. Baimbridge and Simpson (1996) model the financial remuneration of vice chancellors and principals at UK higher institutions using a Times Higher Education Supplement survey. The individual statistical significance of relatively few of their independent variables, together with a large, highly significant, constant term leads them to conclude that managerial and performance indicators fail to offer any explanation for reward levels. They instead establish an idea of a 'going rate' for vice-chancellors. McNabb and Wass (1997) use the census of academic salaries to consider the gender salary gap in academia in 1975, 1985 and 1992. They conclude that women are less successful in achieving promotions from the lecturer scale than their male

counterparts, and receive lower remuneration. Their data, however, lacks variables on individual research productivity.

In contrast the US literature on academic pay, where salary is not determined by a formal pay framework, is extensive, with the main emphasis lying in the investigation of the gender salary gap. Work on the wage tenure profile in academia has been undertaken by Ransom (1993), Brown and Woodbury (1995) and Hallock (1995) who provide some evidence of a negative return to tenure. Ransom (1993) claims that the negative return to tenure is induced by the monopsony power of universities. Johnson and Stafford (1974) and McDowell (1982) consider the effect of career interruption on salary and reveal evidence of negative effects to career breaks within some subjects. US research, however, also generally suffers from the lack of detailed productivity variables. One notable exception is the work by Tuckman, Gapinski and Hagemann (1977) who use cross sectional data from 1972-73 to consider reward to teaching ability, research productivity, public service and administrative skill. They find that research skill. Teaching ability and public service receive small and negligible reward respectively.

Consideration of reservation and deserved wage data has become more usual in economics in recent years. For example, work using data on reservation wages exists in the job-search literature. The most well known of such studies is by Lancaster and Chesher (1983) who use respondent's reservation wages to deduce the structural parameters of the standard optimal job search model. This and many other studies lack a test of the true informational content of data on reservation wages however. One notable exception is provided by Schmidt and Winkelmann (1993), who using a German survey on employed and unemployed individuals and a stationary job search model, compare the stated reservation wages of the unemployed. They find that the two types of reservation wages are consistent with each other, in other words that data on reservation wages are consistent with job search theory. While most previous studies use data on the reservation wages of the unemployed, we use reservation wages of employed. Van den Berg (1992) also uses such data, the major conclusion from his analysis being that moving costs have a substantial impact on the reservation wage.

Work on deserved wages can be found in the job satisfaction literature, where a group of papers have studied comparison effects (see for example, Cappelli and Sherer, 1988; Clark and Oswald, 1996; Hamermesh 1977; Hampton and Heywood, 1999; Sloane and Williams 1996a). Here, workers perceptions of relative or 'comparison' income enters their utility function. Individual job satisfaction is therefore not only affected by a worker's own absolute income level, but also by their income relative to some expected level or comparison group. Hamermesh (1977) concludes that much of the differential in (dis)satisfaction across workers is due to individuals' comparison of their present job with the benchmark opportunities open to them. Although in these studies deserved wage is used as a right-hand-side variable and in our analysis as the left-hand-side variable, this work emphasizes the important informational content of this variable.

3. Data

The data used in this paper come from a unique cross section study of five Scottish Universities: Aberdeen, Dundee, Glasgow, Heriot-Watt and St. Andrews undertaken in 1995/6. The data incorporates detailed information on the personal background, working history, productivity and job satisfaction of 878 academics, collected by means of postal questionnaires.² Academic staff includes professors, senior lecturers and readers, lecturers and research assistants. The overwhelming advantage of this dataset is its uniqueness and detail. It allows us to undertake the first detailed analysis of salary within the UK academic profession. Its comparative disadvantage is it's cross sectional nature. We are only able to analyze a snap-shot of the academic profession at one point in time without the ability to correct for selection in and out of the profession. This restriction is an important caveat to our analysis. Nevertheless the analysis of the cross sectional picture introduces some interesting propositions, to be challenged by future research.

Of the 878 academics from whom information was collected, we select fulltime academics (dropping 48 who work part time), those paid on the non-clinical scale (dropping 51 paid on the clinical scale) and those academics who are under the age of 64 (dropping 3 individuals). The part-time academics are deleted from our sample as we do not have good information on their working hours, which makes the comparison of their wages to the wages of fulltime academics problematic. The academics paid on the clinical scale are dropped due to the difficulty of

incorporating this additional, higher paid, scale in our analysis. From our original sample we also loose 106 observations due to incomplete data and another 35 observations due to intractability of spinal salary point. We are therefore left with 635 observations.

The dataset contains information on an individual's actual, reservation and deserved salary. Actual salary is defined as a respondents's response to the question 'What is your annual salary, that is before any deductions for tax, national insurance, pension contributions, union dues and so on?'. Staff are asked to report this annual salary together with the payment scale of this renumeration. Actual salary therefore refers to pay received on the university payment scale only, that is, it makes unlikely any additional salary attracted from consultancy etc. Reservation salary is questioned though 'What is the lowest salary that you would accept in order to move jobs?'. This question attempts to capture the minimum incentive required for academic mobility, whether it be mobility to another job within the academic profession, or outside. Finally deserved salary information is gathered in response to the question 'In your view, what salary do you deserve to get per annum?' and is questioned in relation to an individuals' current annual salary.

In an attempt to measure the non-pecuniary advantages of an academic position, the dataset also contains detailed information on the advantages and disadvantages of an academic job. Individuals are asked to identify the advantages and disadvantages of an academic career relative to any career alternatives feasible with an individual's present qualifications and experience. Suggested advantages included the flexibility of working timetable, interesting work, the opportunity to travel, a relaxed working environment, the opportunity to teach, geographical mobility and job security. Suggested disadvantages included less supervision/guidance, a closed environment removed from the real world, smaller promotional opportunity, and more limited promotion path. Respondents answer questions on advantages of disadvantage with 'yes' or 'no'.

The definitions of the variables used in our analysis are given in Table 1. Table 2 presents summary statistics for our sample. The first column in table 1 gives the summary statistics for the full sample of 635 observations.³ We see that the majority of Scottish academics are male and are UK citizens. Academics hold on average around 17 years of experience, nearly 10 years

² The average response rate achieved was 30%, reasonably high for this type of study. Data were weighted for non-response at a faculty level by sex allowing for non-response at the level of rank by sex.

³ Descriptive statistics for the complete dataset can be found in Ward (1999).

of which have been spent with their current university. Over 70% of academics hold a PhD and 36% are on short-term contracts. About 31% of our academics are researchers, 34% are lecturers, 21% are senior lecturers or readers and 14% are professors. The science faculty is largest in terms of its staff numbers within the five universities – nearly 40% of academic considered staff work here and Dundee and Glasgow are the largest universities, employing 26% and 32% of our academics respectively. About one out of five respondents are evaluated by their students as a skilled teacher.

Table 2 also presents average statistics on research productivity variables. The average academic has published 20 refereed papers and one book. As research traditions vary substantially by scientific field, table 3 presents these statistics broken down by the faculty that the respondent is working in. The table shows that in the Arts and Social Sciences it is relatively common to write books or chapters in books. On the other hand, the number of published papers is on average substantially higher in Science. The following analysis we take the differences between scientific fields into account by including the number of books, chapters in books, refereed papers, and grants, divided by their averages of the field in which the respondent is working, as explanatory variables.

Table 4 presents the average actual, reservation and deserved salary statistics for academics by rank. We observe that academics across ranks report significantly higher deserved salaries than they actually receive. Staff report underpayment within their current position to the order of 16% for researchers, and around 20% for lecturers, senior lecturers and professors. For researchers and lecturers we observe that the average reservation salary lies between average actual and deserved. Staff in these grades would therefore accept a salary lower that that they felt they deserved in order to move jobs. For senior lecturers, readers and professors, however, average reservation salary is higher than deserved. The staff in these grades are therefore less mobile and would need to be rewarded above the salary they believe they deserve in order to move jobs.

Table 5 gives a matrix representing the number of individuals reporting each of the various combinations between actual, reservation and deserved salary. Points to notice from this table are firstly that the vast majority of respondents report a deserved salary that is greater than that they actually receive. The vast majority also report a reservation salary greater than that they actually receive. This pattern of reports is perhaps what we might expect and might be argued to hold true

for a wider population of workers than academics. There are two interesting, and arguably more unusual, groups that emerge from this matrix however: Academics with low reported deserved salary, that is individuals who report a deserved salary that is either lower or equal to their current salary, and individuals with a low reservation salary, that is individuals who would accept a salary less or equal to their current salary in order to move jobs. Comparison of the majority or 'reference group' and these low deserved and low reservation groups in table 2 reveal more about the average characteristics of these groups. We see that individuals with a low reservation salary are younger than our reference group – they are more likely to be female, in the lower rungs of an academic career and on a short-term contract. We might characterize these workers as the most likely to be mobile. Our low deserved salary individuals are in contrast older, with a higher experience and tenure with current employer. They have had slightly longer periods out of the labour market, and are more likely to be working in St.Andrews, in the faculty of arts or engineering and less likely to work as a lecturer.

Table 6 reports the average response to questions concerning the relative advantages and disadvantages of an academic career. Interesting work and the flexibility of an academic career are the most frequently cited advantages of an academic career over feasible career alternatives. Promotion changes and a less structured promotion path are the most frequently cited disadvantages of academia. Our low reservation academics are more likely to consider academia as removed from the real world. Our low deserved academics value in particular the working environment, interesting work, the flexibility and opportunity to teach within an academic career and are less likely to report promotion related disadvantages of an academic career.

4. The academic salary scale

In the U.K. academic sector, all academic and research staff up to professorial level are paid according to a nationally agreed pay scale. Figure 1 presents the 1994/1995 salary scale.⁴ Academics are placed onto a particular spinal point within a specific scale, such as Lecturer A, by their university and then rise automatically up the rungs or points of a scale, one point each year, until the maximum for that scale is reached. An academic will seek promotion from one

⁴ For a part of our sample the 1994/1995 scale is relevant scale, while for another part the 1995/1996 scale is the relevant scale. Compared to 1994/1995 scale, the salaries of the 1995/1996 scale were increased by 2.7 percent. This fact is taken into account in our analysis.

grade to the next. Accelerated progression up the points of a scale or through the grades and additional salary payments in the form of discretionary awards are possible. There exists however a minimum point at spinal point 4 for those staff with a PhD and a minimum point for individuals aged 27 at spinal point 6. This framework allows us to calculate a minimum spinal point for each academic, on the basis of age, tenure and time-out-of-labour-force. For instance, an academic at age 29 with a tenure of 2 years has to be at least in spinal point 8.

One of the aims of our analysis is to provide some visibility into academic reward through estimates of the returns to individual productivity. In this section we start with the analysis of the determinants of academic salary. The traditional approach is to apply linear regression to a wage equation. However, this approach ignores the data we have on academic positions, and ignores the fact that salaries are not contineously distributed. Furthermore, a wage equation does not allow for policy analysis with respect to changes in the wages within salary scales, or with respect to changes of the salary scale system itself. In order to take account of these problems, we explicitly model the UK academic system of salary scales. Still as a comparison we report and discuss wage regression results.

We observe annual salary and payment scale, such as Lecturer A or B, for all respondents in our sample. Only 26 respondents gave their exact spinal point on the scale. But as several respondents gave an annual salary which fits exactly to a certain point on the salary scale, we can identify a spinal point for an additional 165 respondents. Since there is no formal spinal point system for professors, we model the position of a professor as being spinal point 28. Table 7 outlines the distribution of academics across pay scales. To model the scales and salaries simultaneously, one has to understand that these are outcomes of the same underlying process. As neither of these two kinds of information is perfect, it makes sense to incorporate both pieces of information in a model. To recapulate: the data on the salary scales in not perfect as for the largest part of our sample we only know the respondents' academic position; the data on the wages is not perfect as it clearly contains measurement error.

We model the spinal points and salary scales as an ordered probit, defining x_i as a vector of explanatory variables, β as a parameter vector, and ε_i^s an individual disturbance term. The minimum spinal point that an academic can be in is represented by the point m and the threshold value T^m , which is determined by age, tenure and the time being out of the job.

$$\begin{array}{lll} {(1a)} & {s_i}^* & = {x_i}^{\prime}\beta + {\epsilon_i}^s \\ & s_i & = j & \text{if } T^m \leq T^j < {s_i}^* \leq T^{j+1} \\ & = m & \text{if } & {s_i}^* < T^m \end{array}$$

Next we define w_i as the natural logarithm of the salary of individual i. We model the salaries according to the salary scales of figure 1 with salary w^j for the spinal points j from 4 to 27, with ε_i a individual disturbance term, and $I(s_i = j)$ an indicator function for being on point j.

(1b)
$$w_i = \sum_{j=4,..,27} w^j I(s_i = j) + \varepsilon_i$$

Note that due to the fact that our information on spinal points is imperfect - for most respondents we only observe the salary scale - the salaries contain additional information to estimate the model. In case we would know the exact spinal point for all respondents, equation (1b) would only identify the variance of the error term ε_i – which could be interpreted as measurement error.

Professors are not paid according to the salary scale, and therefore we model their salaries separately. As there is a minimum wage for professors, we model their wages with a censored regression model, with z_i as a vector of explanatory variables, γ as a parameter vector, and ε_i^* as a disturbance term.

(1c)
$$w_i^* = z_i^{\gamma} + \varepsilon_i^*$$

 $w_i^p = w_i^* \quad \text{if } w_i^* > w_i^m$
 $= w_i^m \quad \text{if } w_i^m \ge w_i^*$

Note that the data on the salaries of the professors do not add information to the model of the spinal points, and could be left out of the model. As the salaries of the professors are of interest by themselves however, we include them in our model. For estimation we assume the disturbance terms $(\epsilon_i^s, \epsilon_i, \epsilon_i^*)$ to be independent of the explanatory variables (x_i, z_i) , and to be identically and independently trivariate normally distributed. Our model can be interpreted as an extended version of the switching regression or the Tobit Type 5 model, see Amemiya (1984), in which the switching part of the model is replaced by an ordered probit.

In total, we find that 69 respondents report an annual salary that is below their minimum salary. Although several of these cases might be due to rounding errors in salary, 39 respondents report a wage which is in line with a lower spinal point. Of these 39 respondents, 16 respondents aged 27 or older report a salary that is consistent with a point below the minimal spinal point at age 27. The question is whether this is due to measurement error in our background variables, or whether these individuals really accepted too low a wage. The problem also occurs among the professors - 3 out of 91 professors report a salary that is below the professional minimum of £31,158 in 1994. So although there is an official minimum point, it is an open question whether it is really effective in practice. For the purpose of our analysis, we estimate two models, a model with, and a model without restrictions. In the model with restrictions, we exclude the 39 academics that are paid on, we argue, too low a salary point. Results for this analysis are presented in table 8. Only the constant term differs in significance between the two models.

Table 8 displays the estimation results. We do not include variables such as having a short-term contract and having administrative responsibility among the explanatory variables, since we judge that they are mostly a result of the rank that someone has. This means that they cannot be considered as being exogenous in our model. The model reveals some interesting results. First, we find evidence for some deregulation of established pay and promotion structures; we find that in Heriot-Watt University and in the social sciences academics are put on significantly higher spinal points. This is in line with McNabb and Wass (1997), although contrary to their results we find no significant difference in rewards to full time academics across gender. Second, progression along spinal points is driven almost solely by individual productivity variables. We reveal a positive reward to experience, number of books published, number of refereed papers published, number of grants awarded and high teaching ability⁵. Third, we find evidence of negative effects to career breaks, possibly due to the depreciation effects of career breaks as subject specific skills and knowledge become obsolete.

Our results are in contrast to most of the earlier literature on the gender wage-gap in the academic labour market since the gender-dummy is significant at all conventional significance levels. The reason for this contrasting result might be the fact that we are able to correct for

 $^{^{5}}$ We recognize that there might be some causal effect the other way around – from salary to productivity. None of the papers on this topic mentions this potential problem. And also with our data at hand we see no way to correct for this endogeneity.

productivity. Excluding the productivity variables (books, chapters, papers, grants, having PhD, and teaching skills) from our analysis reveals a significant gender wage-gap at a 10 percent significance level. Most interesting in this respect is the variable out-of-labour-force. Not only are women more likely to have had a out-of-labour-force spell (in our sample 40 percent of women against 10 percent of men), if they have had such a spell the duration is also longer (in our sample 2.5 years on average for women against 1.5 years on average for men). Including an interaction term between gender and the out-of-labour-force time variable reveals that men are not 'punished' significantly differently for such spells to women. The variables experience and tenure do *not* include the out-of-labour-force time, so the results indicate that mothers, and also fathers, who decide to take maternity leave are disadvantaged in the academic labour market. McDowell (1982) argues that durability of knowledge differs significantly per research field. We tested his hypothesis by interacting the out-of-labour-force variable with the faculty variables. We find however no significant differences between the fields of research, which might be due to the fact that the number of observations is small for such a detailed analysis.

Our model of professorial pay is interesting in that none of the explanatory variables included in the model are significant. This suggests that once the position of professor has been attained factors such as experience, publication record and teaching skills are no longer important to reward. Instead one might argue that factors such as negotiation skill, outside offers and costs of moving may be important determinants of professorial pay, which are not captured within our model. This result is in line with Baimbridge and Simpson (1996), who find very few significant variables in their model of the financial remuneration of vice chancellors and principals at UK higher institutions, and instead establish an idea of a 'going rate' for vice-chancellors.

Appendix B presents simulations with respect to productivity and gender-related issues for two reference academics. Although the variables on research productivity is highly significant in our model, the size of their impact turns out to be modest. Remarkle is the impact of teaching skills, for the expected salary the reward to good teaching skills equals the reward to 12 to 15 refereed papers! An explanation for this effect might be that our teaching skill variable picks up other skills, such as presentation skills. Also remarkeble is the impact of out-of-labour-force time; in expected salary a one-year spell has to be compensated with 4 to 5 refereed papers. Although the gender-variable itself is not significant, our simulations show that the impact of gender-related issues might be considerable. Changing our male reference academics without an out-of-labour-

force spell into a woman with a one year out-of-labour-time spell decreases, *ceteris paribus*, the expected salary by 2.8 to 3.8 percent.

As a comparison of our results, we also run a human capital regression of salary against individual characteristics. The detailed results of this are discussed in appendix C. The main finding is that the overall conclusions from this exercise are very much in line with the results from our spinal point and salary scale model.

5. Reservation and deserved salary

In this section we investigate the underlying determinants of academics' reservation and deserved salary. In doing so we hoped to determine those academics most at risk from being lost to the profession. Although mobility in itself is not a bad thing for academics, one would hope that the profession is able to retain the best and most productive academics. We define W_i as an academic's actual annual salary, W_i^r as his/her reservation salary, and W_i^d as his/her deserved salary. x_i is a vector of explanatory variables, β a parameter vector, and ε_i an individual disturbance term. We assume actual salary to be exogenous, and we analyse the deviation of reservation and deserved salary from actual salary using seemingly unrelated regression:

(2a)
$$100 (W_i^r - W_i) / W_i = x_i^{3} \beta^r + \epsilon^r$$

(2b)
$$100 (W_i^d - W_i) / W_i = x_i^{\beta d} + \epsilon^d$$

We regress the percentage deviation of reservation and deserved salary from actual salary on the same set of explanatory variables utilised in section 4. One could also argue that non-pecuniary advantages and disadvantages of the job may be important in the determination of reservation and deserved wages. This information is therefore also included as a series of dummy variables. Results are presented in table 9. The significance of the correlation coefficient suggests that there are unobserved variables that determinate both academic's reservation and deserved salary.

Considering reservation wages first, we see that *ceteris paribus* the impact of salary is u-shaped with the minimum at the top of the Lecturer B scale with an annual salary of £26,574. Although

hardly significant, experience is n-shaped with the maximum at 2.3 years of experience. The insignificant results on the productivity variables in table 9 provide a neutral answer to our question concerning whether academia can retain its most productive staff. It appears that good academics are at least not setting low reservation wages for themselves in order to leave the profession. On the other hand, this is also true of the less productive academics. A surprising result is the impact of the number of chapters, which has a significantly negative impact! In combination with the insignificant impact of the number of chapters in the salary scale model, a reasonable explanation seems to be that publishing chapters in books is under-valued in British academia. Finally we find only weakly significant impacts of the non-pecuniary factors of geographical mobility and being removed from reality. Overall, therefore, pecuniary considerations dominate the determination of academic's reservation wage.

Turning to the results for deserved salary again reveals a seniority effect of higher wages on the percentage deviation of deserved salary from actual salary. This time the impact of wages is insignificant but the minimum point of the wage-squared function lies at £38.571 – within the professorial grade. The effect of experience is significant and n-shaped with the maximum at an experience of 23 years. Those with less experience are increasingly discontented with their appointed salary point. Alternatively, this result may reflect a selection effect. Those with a lot of experience within the profession may be those who have achieved their best match. Staff with less experience may not yet have done so, and it is possible that some of these less satisfied individuals leave the profession. The 'correct' interpretation may also dependent on the reference group referred to by the respondent within the assessment of deserved salary. For young, more recently qualified academics the relevant reference group may be more likely to be other young workers, including those working in the private sector who hold a comparative wage advantage. On the other hand, older academics holding mainly specialised human capital may compare themselves with their peers within the profession.

Also striking within this analysis are the results on the skilled teacher variable, the size of which is considerable at 6.7 percent. Thus, although we find evidence of a significant reward to teaching skill in the analysis of actual salary, this reward is insufficient in the eyes of the academics themselves. We find no evidence of academics perceiving penalties to time out of labour force. The effect of this variable on deserved wage is insignificant. Finally, the results concerning the (dis-)advantages of academia identify more about the grievances of our most dissatisfied academics than any evidence of positive compensating effects. Dissatisfaction with promotional prospects increases deserved salary demands substantially. This result suggests the interdependence between salary and position within the formal academic reward system. Perceptions of academia as being removed from the real world reduces deserved salary.

6. Conclusion.

Our analysis of actual, reservation and deserved salary within the UK academic profession has uncovered a number of interesting effects. Firstly, our results outline the importance of publication, grant receipt and teaching skill in attracting financial reward within the current payment system. With the inclusion of the salary framework into analysis the relationship between productivity and individual reward is reinforced. Our model therefore provides some evidence to ease concerns that the current reward system does not adequately recognise individual productivity. Perhaps surprising in this respect is the importance of teaching skills, which are revealed to have a sizable impact on pay in our simulations. Secondly, we find some suggestion of a negative reward to time out of the profession - career breaks carry an associated penalty, perhaps due to depreciation effects as subject specific skills and knowledge become obsolete. Simulations show that this effect is sizable; the salary loss associated with a one-year out-of-labour-force spell would require compensation equivalent to 4 to 5 additional refereed papers. Third, none of the explanatory variables included in our model of professorial pay are significant. This suggests that once the position of professor has been attained factors such as experience, publication record and teaching skills are no longer important to reward. Instead factors such as negotiation skill, outside offers and costs of moving may be important determinants of professorial pay.

Analysis of deserved and reservation salary suggests that controlling for individual characteristics, lecturers hold the lowest reservation salaries in relation to their current salary level. The profession is therefore most at risk from loosing its staff at this grade– presumably those are the ones who will find it easiest to attract job offers from outside academia. We find that professors are most at ease with their salary position. In contrast, lower rung academics are the least content. Analysis reveals some evidence of dissatisfaction with actual pay for those with high teaching ability, but we find no (self-)selection on the basis of individual productivity.

References

Amemiya, T. (1984), Tobit Models: A Survey, Journal of Econometrics, Vol. 24, pp. 3-61.

Baimbridge, M. and Simpson, C. (1996), Rewards to Academia: the remuneration of vice chancellors and Principals, *Applied Economics*, Vol. 28, pp.631-639.

Brown, B. W. and Woodbury, S. A. (1995), Seniority, External Labor Markets and Faculty Pay, *Staff Working Paper 95-37*, Upjohn Institute for Employment Research.

Cappelli, P. and Sherer, P. (1988) Satisfaction, Market Wages and Labor Relations: An Airline Study, Industrial Relations 27, pp. 556-73.

Clark, A. E. and Oswald, A. J. (1996), 'Satisfaction, and Comparison Income', *Journal of Public Economics*, Vol. 61, No. 3, September pp. 359-382.

Dearing (1997) National Committee of Inquiry into Higher Education, *Higher Education in the Learning Society*, Report of the National Committee, Main Report, HMSO.

Hamermesh, D. S. (1977), 'Economic Aspects of Job Satisfaction, in Ashenfelter, O.E and Oates, W.E, editors, *Essays in Labor Market Analysis*, John Wiley, New York, pp.53-72.

Hampton, M., Heywood, J. (1999), The Determinants of Perceived Underpayment: The Role of Racial Comparisons, Review of Social Economy 57, pp.141-155.

Hay (1997), *A study of pay and conditions in universities and colleges of Higher Education*, report by the Independent Pay Commission.

Hallock, K. F. (1995), Seniority and Monopsony in the Academic Labour Market: Comment, *American Economic Review*, Vol. 85, No. 3, pp.654-57.

Johnson, E and Stafford, F. P. (1974), The Earnings and Promotion of Women Faculty, *The American Economic Review*, Vol. 64, No. 6, pp.888-903.

Lancaster, T. and Chesher A. (1983), An Econometric Analysis of Reservation Wages, *Econometrica*, Vol. 51, pp. 1661-1676.

McDowell, J. (1982), Obsolescence of Knowledge and Career Publication Profiles: Some Evidence of Differences among Fields in Costs of Interrupted Careers, *The American Economic Review*, Vol. 72, No. 4, pp.752-768.

McNabb, R. and Wass V. (1997), Male-female salary differentials in British universities, *Oxford Economic Papers*, Vol. 49, Issue 3, pp. 328-343.

Ransom, M. R. (1993), Seniority and Monopsony in the Academic Labor Market, *The American Economic Review*, Vol. 83, No. 1, pp.221-233.

Schmidt, C., and Winkelmann R. (1993), Reservation Wages, Wage Offer Distribution, and Accepted Wages, in Bunzel, et al. (eds.), *Panel Data and Labour Market Dynamics*, Elsevier Science Publisher.

Sloane, P. J and Williams, H. (1996a), 'Are "Overpaid" Workers really unhappy? A Test of the Theory of Cognitive Dissonance' *Labour*, Vol. 10, No. 1, pp.3-15.

Tuckman, H. P., Gapinski, J. H. and Hagemann, R. P. (1977), Faculty Skills and the Salary Structure in Academe: A Market Perspective, *The American Economic Review*, Vol. 67, No. 4, pp.692-702.

Van den Berg, G. J. (1992), A Structural Dynamic Analysis of Job Turnover and the Costs Associated with Moving to Another Job, *The Economic Journal*, Vol. 102, pp. 1116-1133.

Ward, M. (1999), Your Everyday, Average Academic, IZA DP, No. 63.

Ward, M. and Sloane P. J. (1999), Non-Pecuniary Advantages versus Pecuniary Disadvantages; Job Satisfaction among Male and Female Academics in Scottish Universities, *Scottish Journal of Political Economy*, forthcoming.

SALARY SCALES 1994/1995

Joint Negotiating Committee for Non-Clinical Academic and Academic Related Staff

Spinal Point	Salary on 1 April 1994	ļ	ACAE	DEMI	С	RES	EAR	СН	ST	AFF
4	13,941					E				#
5	14,756					RAD	9			
6	15,566	-				ß				*
7	16,191	ER /							≤	
8	17,007	UR							DEI	
9	17,813	ECI						I	IRA.	
10	18,486	Γ						I	0	
11	19,326							Ĩ		
12	20,133									
13	20,953									
14	21,786									
15	22,622	۶B				Π				
16	23,498	RE				DE				
17	24,377	CTU				GRA				
18	25,735	LE								
**20	27,018		ry				ry	Í		
21	27,881		Points	2			Points		H	
22	28,756		Ö	La L			Di		DEI	
23	29,646			LUH				Ĩ	RAI	
24	30,533							[G	
25	31,302									
26	32,094			SF 1						
27	33,007									

Professorial Minimum £31,158 Grade IV Minimum £31,158

Notes: * Age 27 point

Minimum appointment level for staff with PhD ** Point 19 was deleted with effect from 1.4.91

Table 1: list of variables

Name	Definition
Individual char.	
Gender	,= 1 if male, = 0 if female
Citizen	,= 1 if UK citizen, = 0 otherwise
Job char.	
Experience	Lenght total labour market experience, measured in years ¹
Job tenure	Lenght of time with current employer, measured in years ¹
Time-out	Lenght of time out of labour force, measured in years
Short-term	,= 1 if contract for 3 or less years, = 0 otherwise
Job position	
Researcher	,= 1 if researcher, = 0 otherwise
Lecturer	,= 1 if lecturer A or lecturer B, = 0 otherwise
Sen.Lec./Reader	,= 1 if senior lecturer or reader, = 0 otherwise
Professor	,= 1 if professor, = 0 otherwise
<u>University</u>	Dummies for the five universities from which the data are sampled from
Faculty	Dummies for the five faculties from which the data are sampled from
Publications	
Books	Total number of books published
Chapters	Total number of chapters published in books
Papers	Total number of refereed publications published
<u>Other</u>	
Grants	Total number of grants received
Having PhD	,= 1 if holds a PhD, = 0 otherwise
Teach.skill	,= 1 if skilled teacher (based on student's evaluations), = 0 otherwise
Advantages	Advantages of present career over feasible alternatives according to respondent's opinion
Environment	,= 1 if relaxed working environment is an advantage, = 0 otherwise
Interesting work	,= 1 if interesting work is an advantage, = 0 otherwise
Responsibility	,= 1 if opportunity to hold responsibility is an advantage, = 0 otherwise
Job Security	,= 1 if job safety is an advantage, = 0 otherwise
Flexibility	,= 1 if flexible working timetable is an advantage, = 0 otherwise
Traveling	,= 1 if opportunity to travel is an advantage, = 0 otherwise
Mobility	,= 1 if geographical mobility is an advantage, = 0 otherwise
Teaching	,= 1 if opportunity to teach is an advantage, = 0 otherwise
<u>Disadvantages</u>	Disadvantages of present career over feasible alternatives according to respondent's opinion
Supervision	,= 1 if less supervision is a disadvantage, = 0 otherwise
Prom. Chances	,= 1 if smaller opportunity for promotion is a disadvantage, = 0 otherwise
Prom. Path	,= 1 if more limited promotion path is a disadvantage, = 0 otherwise
Reality	,= 1 if removal from real world is a disadvantage, = 0 otherwise

¹⁾ experience and tenure do not include the time being out of the labour force.

	Full sar	nple	Reference	group	Low res.	wage	Low des. V	Vage
			w ^r > w, w	^d > w	W ^r <= 7	w	w ^d <= 7	w
	(635 o	bs.)	(282 ol	os.)	(175 ob	os.)	(97 obs	s.)
Individual char.								
Age -29	0.213		0.202		0.303		0.216	
Age 30-39	0.312		0.319		0.367		0.247	
Age 40-49	0.265		0.270		0.240		0.289	
Age 50-	0.211		0.209		0.091		0.247	
Gender	0.691		0.734		0.611		0.701	
Citizen	0.882		0.862		0.903		0.928	
<u>Job char.</u>								
Experience	16.731	(10.854)	16.668	(10.379)	13.243	(9.978)	17.856	(11.150)
Job tenure	9.700	(9.997)	9.790	(10.052)	6.864	(7.696)	10.077	(10.473)
Time-out	0.422	(1.646)	0.341	(1.302)	0.471	(1.457)	0.672	(2.869)
Short-term	0.361		0.323		0.571		0.351	
Job position								
Researcher	0.312		0.284		0.474		0.351	
Lecturer	0.340		0.355		0.326		0.206	
Sen.Lec./Reader	0.205		0.202		0.114		0.258	
Professor	0.143		0.160		0.086		0.186	
University								
Aberdeen	0.162		0.167		0.120		0.103	
Dundee	0.260		0.241		0.263		0.299	
Heriot-Watts	0.068		0.078		0.069		0.041	
St.Andrews	0.192		0.184		0.223		0.237	
Glasgow	0.318		0.330		0.326		0.320	
Faculty								
Arts	0.170		0.131		0.166		0.237	
Engineer	0.139		0.145		0.086		0.175	
Medicine	0.143		0.160		0.154		0.144	
Science	0.387		0.404		0.457		0.320	
Soc.science	0.161		0.160		0.137		0.124	
Publications								
Books	1.074	(2.418)	1.043	(2.559)	0.783	(2.122)	1.051	(1.856)
Chapters	2.805	(6.515)	2.691	(4.982)	1.800	(3.883)	2.278	(4.361)
Papers	20.109	(28.292)	20.411	(25.966)	14.720	(28.317)	20.422	(30.367)
<u>Other</u>								
Grants	4.951	(8.151)	5.557	(8.069)	3.589	(7.309)	4.463	(8.713)
Having PhD	0.728		0.748		0.709		0.701	
Teach.skill	0.198		0.209		0.149		0.175	

Table 2: sample statistics

Note: the first column presents the statistics of the full sample, the second column represents the statistics of the reference group (both reservation salary w' and deserved salary w' are larger than the actual salary w). Between parentheses the standard deviations.

	number										
Faculty	of obs.	ag	е	boo	ks	cha	pters	рар	bers	gra	nts
Arts	107	45.73	(9.84)	2.28	(3.39)	3.89	(5.22)	13.03	(15.89)	2.54	(4.49)
Engineer	88	39.00	(10.49)	0.28	(0.80)	2.05	(10.90)	16.68	(24.26)	4.48	(6.57)
Medicine	91	36.85	(9.13)	0.46	(1.11)	2.22	(4.51)	19.68	(26.27)	6.12	(10.86)
Science	246	39.42	(10.50)	0.60	(1.38)	2.13	(5.06)	26.50	(34.58)	6.13	(9.39)
Soc.Science	103	40.80	(10.35)	2.17	(3.74)	4.49	(6.74)	15.50	(23.40)	4.02	(5.28)

Table 3: research productivity statistics

Note: Standard deviations between parentheses.

Table 4: salary statistics

	number of Actual			Reserva	ation	Deserved		
	observations	salary		salar	у	salary		
Researcher	162	17.24	(2.66)	18.04	(5.02)	19.97	(4.57)	
Lecturer	154	22.20	(3.55)	25.36	(7.33)	26.85	(5.53)	
Sen.Lec/Reader	84	29.98	(1.83)	37.02	(11.43)	35.63	(5.11)	
Professor	66	37.30	(4.01)	47.56	(18.96)	44.61	(8.47)	

Note: only observations with actual, reservation, and deserved salary observed are included. Salary in 1,000 BP per year, and comparable to the 1994/1995 salary scale. Standard deviations between parentheses.

Table 5: observations on salary

	w ^d missing	w ^d < w	$w^d = w$	$w^d > w$	Total
w ^r missing	34	8	20	53	115
w ^r < w	11	18	17	76	122
w ^r = w	6	3	5	39	53
$w^r > w$	37	5	21	282	345
Total	88	34	63	450	635

Table 6: (dis-)advantages statistics

	Full sample	Reference group	Low res. wage	Low des. wage
	(635 obs.)	(282 obs.)	(175 obs.)	(97 obs.)
<u>Advantages</u>				
Environment	0.516	0.475	0.549	0.598
Interesting work	0.885	0.922	0.817	0.928
Responsibility	0.403	0.422	0.326	0.381
Job Security	0.313	0.305	0.269	0.278
Flexibility	0.789	0.805	0.783	0.814
Travelling	0.529	0.589	0.497	0.526
Mobility	0.100	0.106	0.126	0.113
Teaching	0.498	0.493	0.451	0.536
Disadv.				
Supervision	0.081	0.064	0.131	0.103
Prom. Chances	0.457	0.532	0.463	0.299
Prom. Path	0.469	0.514	0.497	0.278
Reality	0.170	0.128	0.280	0.237

Note: the first column presents the statistics of the full sample, the second column represents the statistics of the reference group (both reservation salary w^r and deserved salary w^d are larger than the actual salary w). Between parentheses the standard deviations.

Sp	pinal point	Academic staff	Research staff
4	(3 obs.)		Grade I.B
5	(12 obs.)		Points 4-6
6	(18 obs.)		(26 obs.)
7	(9 obs.)	Lecturer A	
8	(9 obs.)	Points 5-11	Grade I.A
9	(14 obs.)	(58 obs.)	Points 4 –13
10	(7 obs.)		(77 obs.)
11	(11 obs.)		
12	(9 obs.)		
13	(10 obs.)		
14	(2 obs.)		
15	(7 obs.)	Lecturer B	Grade II
16	(3 obs.)	Points 12-22	Points 11-22
17	(6 obs.)	(88 obs.)	(13 obs.)
18	(18 obs.)		
20	(11 obs.)		
21	(7 obs.)		
22	(3 obs.)		
23	(2 obs.)	Sen.Lec./Reader	Grade III
24	(15 obs.)	Points 20-27	Points 17-27
25	(8 obs.)	(89 obs.)	(2 obs.)
26	(4 obs.)		
27	(3 obs.)		
		Professor	
		(91 obs.)	

Table 7: observations on spinal points and salary scales

	I	Model without	restrictions		Ν	lodel with rest	rictions	
	Spin	al Point	Ln(S	alary Prof.)	Spin	al Point	Ln(S	alary Prof.)
	Par.	s.e.	par.	s.e.	par.	s.e.	par.	s.e
Individual char.								
Intercept	0.4076	(0.3799)	10.597	***(0.1312)	-0.4523	(0.5174)	10.5917	***(0.1085)
Sex	0.1109	(0.1385)			0.1362	(0.1794)		
Citizen	0.1266	(0.1763)			0.2781	(0.2313)		
Job Char.								
Experience	0.1929	***(0.0271)			0.2405	***(0.0345)		
Experience ² /10	-0.0274	***(0.0068)			-0.0358	***(0.0081)		
Tenure	0.0201	(0.0268)	0.0036	(0.0058)	-0.0288	(0.0323)	0.0038	(0.0056)
Tenure ² /10	-0.0054	(0.0086)	-0.0018	(0.0017)	0.0075	(0.0100)	-0.0018	(0.0017)
Time-out	-0.0952	***(0.0351)			-0.1201	**(0.0485)		
<u>University</u>								
Aberdeen	0.1704	(0.1768)	-0.0188	(0.0446)	0.1981	(0.2030)	-0.0194	(0.0437)
Dundee	0.0445	(0.1526)	-0.0044	(0.0575)	0.0899	(0.1962)	-0.0054	(0.0565)
Heriot-Watts	0.5524	***(0.2579)	-0.0614	(0.0587)	0.6011	**(0.2973)	-0.0607	(0.0574)
St.Andrews	-0.0655	(0.1709)	0.0612	(0.0460)	-0.0395	(0.2109)	0.0621	(0.0451)
Faculty								
Arts	0.1021	(0.1749)	-0.0102	(0.0487)	0.1064	(0.2052)	-0.0112	(0.0478)
Engineer	-0.2895	(0.1905)	0.0470	(0.0578)	-0.3974	(0.2423)	0.0502	(0.0567)
Medicine	-0.0207	(0.1904)	0.0768	(0.0613)	-0.0758	(0.2362)	0.0076	(0.0602)
Soc.Science	0.3611	**(0.1859)	-0.0143	(0.0505)	0.3070	(0.2224)	-0.0131	(0.0485)
Publications								
Books / average by fac.	0.1165	***(0.0369)	-0.0060	(0.0059)	0.1223	***(0.0397)	-0.0056	(0.0059)
Chapt. / average by fac.	-0.0216	(0.0254)	-0.0048	(0.0055)	-0.0240	(0.0268)	-0.0050	(0.0054)
Papers / average by fac.	0.3614	***(0.0747)	0.0009	(0.0114)	0.3933	***(0.0815)	0.0008	(0.0108)
Other								
Grants / average by fac.	0.1333	***(0.0462)	0.0064	(0.0101)	0.1616	***(0.0502)	0.0060	(0.0100)
Having PhD	0.2323	(0.1423)	-0.0434	(0.0441)	-0.0279	(0.1788)	-0.0390	(0.0414)
Teach.skill	0.3229	**(0.1527)	0.0384	(0.0359)	0.3208	*(0.1717)	0.0374	(0.0351)
Distr. parameters								
Standard dev.	0.0728	(0.0047)	0.0928	(0.0122)	0.0721	(0.0047)	0.0908	(0.0116)
Correlation	-0.5043	(0.0717)	-0.0213	(0.5083)	-0.4487	(0.0896)	-0.2134	(0.4729)

Table 8: Estimation results on spinal point and salary scale

	Reserv	ation salary	Deser	ved salary
	100(W	-W)/W	100(W	^d -W)/W
	par.	s.e.	par.	s.e.
Individual char.	F -			
Intercept	46.0216	***(19.8374)	28.7897	***(11.0619)
Gender	-0.7159	(3.4582)	-0.5066	(1.9284)
Citizen	-2.6702	(4.2539)	-0.1478	(2.3721)
Job char.				
(W/10,000)	-37.2322	***(15.6570)	-13.5078	(8.7308)
(W/10,000) ²	7.0052	***(2.5820)	1.7516	(1.4398)
Experience	1.1957	*(0.6868)	0.7518	**(0.3830)
Experience ² /10	-0.2621	(0.1767)	-0.1577	(0.0985)
Tenure	0.3020	(0.6604)	0.0478	(0.3683)
Tenure ² /10	0.1234	(0.2139)	0.0452	(0.1193)
Time-out	-1.0188	(1.0803)	-0.7616	(0.6024)
University		, , , , , , , , , , , , , , , , , , ,		, , , , , , , , , , , , , , , , , , ,
Aberdeen	1.5043	(4.4223)	2.1708	(2.4660)
Dundee	1.0357	(3.7018)	-2.0651	(2.0642)
Heriot-Watts	-0.0763	(5.9366)	0.8174	(3.3104)
St.Andrews	4.0506	(4.0109)	-0.2100	(2.2366)
Faculty		, , , , , , , , , , , , , , , , , , ,		, , , , , , , , , , , , , , , , , , ,
Arts	-6.7891	(4.3780)	-3.3358	(2.4413)
Engineer	5.3442	(4.6596)	-1.2326	(2.5983)
Medicine	-0.5462	(4.3939)	-1.0640	(2.4502)
Soc.Science	-0.8213	(4.4326)	-1.9342	(2.4717)
Publications		, , , , , , , , , , , , , , , , , , ,		, , , , , , , , , , , , , , , , , , ,
Books / average by fac.	0.6081	(0.7685)	0.0474	(0.4286)
Chapt. / average by fac.	-2.0800	**(0.9370)	-0.9156	*(0.5225)
Papers / average by fac.	2.2234	(1.4724)	0.9480	(0.8210)
Other				
Grant / average by fac.	-1.1425	(1.2726)	0.3022	(0.7097)
Having PhD	-1.2046	(3.4186)	3.4236	*(1.9063)
Teach.skill	1.9112	(3.7400)	6.6908	***(2.0855)
Advantages				
Environment	0.2705	(2.9771)	-1.3194	(1.6601)
Interesting work	5.9962	(4.7202)	0.9330	(2.6321)
Responsibility	2.3727	(3.0073)	2.4883	(1.6769)
Job Security	0.0771	(3.3258)	0.7457	(1.8545)
Flexibility	-5.2465	(3.7554)	-1.4879	(2.0941)
Traveling	4.0951	(2.9156)	0.2794	(1.6258)
Mobility	-8.1788	*(4.4352)	-0.1649	(2.4732)
Teaching	1.0688	(3.0293)	-2.2623	(1.6892)
<u>Disadvantages</u>				
Supervision	-7.7305	(4.9172)	-3.1455	(2.7420)
Prom. Chances	3.1738	(3.1769)	4.0241	**(1.7715)
Prom. Path	-2.9296	(3.0845)	3.4224	**(1.7200)
Reality	-6.5586	*(3.6697)	-4.6591	**(2.0463)
Correlation coeff.		0.2	2664***	

Table 9: Estimation results on reservation and desired salary

Appendix A: Estimation of spinal point and salary scale model

We model the spinal points and salary scales as an ordered probit. Define x_i as a vector of explanatory variables, β as a parameter vector, and ε_i^s an individual disturbance term. The minimum spinal point that an academic can be in is represented by the point m and the treshold value T^m , which is determined by age, tenure and the time being out of the job.

$$\begin{array}{lll} {(1a)} & {s_i}^* & = {x_i}^{\prime}\beta + {\epsilon_i}^s \\ & s_i & = j & \quad \mbox{if } T^m \leq T^j < {s_i}^* \leq T^{j+1} \\ & = m & \quad \mbox{if } & {s_i}^* < T^m \end{array}$$

Next define w_i as the natural logarithm of the salary of individual i. We model the salaries according to the salary scales of figure 1 with salary w^j for the scales j from 4 to 27, with ε_i a individual disturbance term, and $I(s_i = j)$ an indicator function for being in scale j.

(1b)
$$w_i = \sum_{j=4,..,27} w^j I(s_i = j) + \varepsilon_i$$

Note that due to the fact that our information on spinal points is imperfect - for most respondents we only observe the salary scale - the salaries contain additional information to estimate the model. In case we would know the exact spinal point for all respondents, equation (1b) would only identify the variance of the error term ε_i – which could be interpreted as measurement error.

Professors are not paid according to the salary scale, and therefore we model their salaries separately. As there is a minimum wage for professors, we model their wages with a censored regression model, with z_i as a vector of explanatory variables, γ as a parameter vector, and ε_i^* as a disturbance term.

(1c)
$$w_i^* = z_i^2 \gamma + \varepsilon_i^*$$

 $w_i^p = w_i^* \quad \text{if } w_i^* > w_i^m$
 $= w_i^m \quad \text{if } w_i^m \ge w_i^*$

Note that the data on the salaries of the professors do actually not add information to the model of the spinal points, and could be left out of the model. As the salary equation of the professors is interesting in itself, we include it in our model. For estimation we assume that the disturbance terms ($\epsilon_i^s, \epsilon_i, \epsilon_i^*$) to be independent of the explanatory variables (x_i, z_i) and to be identically and independently trivariate normally distributed. Our model can be interpreted as an extended version of the switching regression or the Tobit Type 5 model, see Amimiya (1984), in which the switching part of the model is replaced by an ordered probit. As from the data it is not clear whether the restrictions on the scales and the professional salaries hold in practice, we decide to estimate one model without restrictions, and one model with restrictions. We first discuss the estimation of the model without the restrictions.

Model without restrictions

The likelihood contribution for an academic i in scale j and wage w_i is:

$$P(s_{i}=j, w_{i}) = P(s_{i}^{*} \le T^{j+1}, w_{i}) - P(s_{i}^{*} \le T^{j}, w_{i})$$

=
$$[P(s_{i}^{*} \le T^{j+1} | w_{i}) - P(s_{i}^{*} \le T^{j} | w_{i})] P(w_{i})$$

with for non-professors (j \leq 27): P($s_i^* \leq T^j | w_i$) = $\Phi((T^j - x_i^*\beta - (\rho/\sigma)(w_i - w^j)) / \sqrt{(1-\rho^2)})$ P(w_i) = $\phi((w_i - w^j) / \sigma)$

and for professors (j=28): P($s_i^* \leq T^j | w_i$) = $\Phi((T^j - x_i^*\beta - (\rho^*/\sigma^*)(w_i - z_i^*\gamma)) / \sqrt{(1-\rho^{*2})})$ P(w_i) = $\phi((w_i - z_i^*\gamma)/\sigma^*)$

Note that the standard deviation of ε_i^s is set to one. Note also that for this model the correlation between the error-terms of the wages for the non-professors and professors is not identified. For a comparable result, see the Tobit Type 5 model of Amemiya (1984).

Model with restrictions

The likelihood contribution for a non-professor i in scale j and wage w_i is:

$$P(s_i=j, w_i | s_i \ge m) = P(s_i=j, w_i) / P(s_i \ge m)$$
 (j≥m)

As the nominator is the same as for the model without restrictions, the derivation of the likelihood contribution is furthermore straightforward. For estimation we deleted the 39 individuals with j < m from the data. The likelihood contribution for a professor i with wage w_i is:

$$P(s_i=28, w_i | s_i \ge m, w_i \ge w_i^m) = P(s_i=28, w_i) / P(s_i \ge m, w_i \ge w_i^m)$$

Again the nominator is the same as for the model without restrictions. For 3 professors with a reported salary below the professional minimum, we set the salary equal to this professional minimum. Note that for this model the correlation between the non-professional and the professional wage is identified. Still the maximum likelihood procedure (of GAUSS) has problems to optimise the likelihood with respect to this parameter. As this parameter is only identified on the basis of the data on the professors, and the minimum scale restriction is of little importance for the professors, this is not a surprise. We set this correlation equal to zero.

Appendix B: Simulation results

This appendix discusses the simulations of our salary scale model. For the salary scale model we present the results for the model with restrictions. Results should therefore be interpreted as an upper bound, since the model is based on a sample excluding certain academics (see discussion following equation 1c, section 4). As reference academics we choose two academics in social sciences; one academic with characteristics close to the average lecturer, and one academic with characteristics close to the average senior lecturer/reader. Tables B.1 and B.2 present the characteristics of these reference academics. In our simulation we do not restrict outcomes to particular spinal points, but allow them instead to vary. Besides calculating the probabilities according to differing spinal points and the resulting expected wage for the reference academics,

we also calculate the impact on salary of an additional book, chapter, paper, grant, teaching skills, 1 year out-of-labour-force, gender, and gender plus 1 year out-of-labour-force.

We first discuss the simulation of the reference academics in tables B.1 and B.2. Table B1 shows a lot of probability mass at spinal points 11 and 18, which most likely represents clustering at the top of the lecturer A and lecturer B scales as individuals wait for promotion to the next salary scale. We also find a high mass in points 12, 13 and 20. In table B.2 the relevant high mass points are 18, 20, 24 and being a professor. The predicted salaries seem in line with what might be expected on the basis of the actual salaries in the data.

Although the variables on research productivity were highly significant in our model, the size of their impact is modest in our simulations. An additional published book increases the expected wage by 0.7 to 0.8 percent, while an additional published paper increases the expected wage by 0.3 to 0.4 percent. Astonishing in this respect is the impact of teaching skills, a change to having good teaching skills increases the expected salary by 4.1 to 6.2 percent. For our second reference academic this change increases the probability of being professor from 15.1 to 23.8 percent. An explanation for this large effect is that the teaching skill variable also picks up other skills, like presentation skills. Also large is the impact of an out-of-the-labour-force spell, a one-year spell decreases the expected salary by 1.3 to 1.8 percent. Although gender itself was not significant in our model, the simulations shows that the impact of gender-related issues might be considerabele. Changing our male reference academics without an out-of-labour-force spell into woman with a one year out-of-labour-force spell decreases, *ceteris paribus*, the expected salary by 2.8 to 3.8 percent.

reference	Male citizen, 36 years, 12 years experience, 4 years tenure, no time-out-of-labour-force										
academic 1 =	1	book, 2 ch	apters, 6 pap	ers, 2 grants	s, PhD, no te	eaching skil	ls				
Scale	ref.	+1 book	+1 chapt.	+1 paper	+1 grant	+tch.sk.	+1 out-of	woman	+1 out-of		
4	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00		
5	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00		
6	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00		
7	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00		
8	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00		
9	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00		
10	4.26	3.99	4.29	4.14	4.07	2.85	4.87	4.95	5.61		
11	13.89	13.17	13.95	13.56	13.38	10.03	15.44	15.65	17.24		
12	10.38	10.00	10.41	10.21	10.11	8.18	11.16	11.27	12.01		
13	15.85	15.51	15.88	15.70	15.61	13.63	16.50	16.58	17.12		
14	2.06	2.03	2.06	2.05	2.04	1.88	2.10	2.10	2.12		
15	7.44	7.39	7.44	7.42	7.41	7.00	7.49	7.49	7.48		
16	3.28	3.28	3.27	3.28	3.28	3.19	3.25	3.25	3.20		
17	7.62	7.67	7.61	7.64	7.66	7.72	7.45	7.43	7.21		
18	14.34	14.67	14.31	14.50	14.58	15.89	13.59	13.48	12.66		
20	10.10	10.56	10.06	10.31	10.43	12.67	9.13	9.00	8.07		
21	4.03	4.29	4.01	4.15	4.22	5.62	3.50	3.44	2.96		
22	1.45	1.56	1.44	1.50	1.53	2.15	1.23	1.21	1.02		
23	0.61	0.66	0.61	0.63	0.65	0.93	0.51	0.50	0.42		
24	3.21	3.52	3.18	3.35	3.43	5.26	2.63	2.55	2.07		
25	0.73	0.81	0.72	0.77	0.79	1.35	0.57	0.55	0.42		
26	0.20	0.22	0.19	0.21	0.21	0.39	0.15	0.14	0.11		
27	0.10	0.11	0.10	0.11	0.11	0.20	0.08	0.07	0.05		
Prof.	0.47	0.54	0.46	0.50	0.52	1.06	0.34	0.33	0.23		
<u>Salary</u>											
in £1,000	23.26	23.42	23.25	23.33	23.37	24.21	22.95	22.91	22.62		
Deviation											
Percent	0.00	0.65	-0.06	0.30	0.47	4.05	-1.34	-1.51	-2.76		

Table B.1: simulations on individual and productivity characteristics (1)

Reference	male citizen, 48 years, 24 years experience, 16 years tenure, no time-out-of-labour-force								
academic 2 =	2	2 books, 4 c	hapters, 16 p	oapers, 6 gra	ants, PhD, n	o teaching s	skills		
scale	ref.	+1 book	+1 chapt.	+1 paper	+1 grant	+ tch.sk.	+1 out-of	woman	+1 out-of
4	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00
5	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00
6	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00
7	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00
8	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00
9	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00
10	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00
11	1.17	1.03	1.19	1.11	1.07	0.55	1.52	1.57	2.01
12	1.36	1.22	1.38	1.29	1.26	0.69	1.71	1.76	2.18
13	3.25	2.96	3.28	3.12	3.04	1.81	3.95	4.05	4.85
14	0.56	0.52	0.57	0.54	0.53	0.33	0.67	0.69	0.80
15	2.39	2.21	2.41	2.31	2.26	1.45	2.82	2.88	3.33
16	1.27	1.18	1.27	1.22	1.20	0.80	1.47	1.50	1.71
17	3.60	3.36	3.62	3.49	3.43	2.35	4.11	4.18	4.71
18	10.73	10.20	10.78	10.49	10.35	7.70	11.85	12.00	13.05
20	14.28	13.87	14.32	14.09	13.99	11.62	15.05	15.14	15.72
21	9.65	9.54	9.66	9.60	9.58	8.69	9.79	9.80	9.80
22	4.68	4.68	4.68	4.68	4.68	4.46	4.65	4.64	4.54
23	2.27	2.28	2.27	2.27	2.28	2.22	2.23	2.22	2.15
24	17.85	18.18	17.82	18.01	18.09	19.04	17.00	16.88	15.84
25	7.50	7.80	7.47	7.64	7.71	8.99	6.83	6.74	6.05
26	2.72	2.85	2.70	2.78	2.82	3.45	2.42	2.38	2.10
27	1.59	1.68	1.58	1.63	1.65	2.07	1.40	1.38	1.20
Prof.	15.12	16.46	14.99	15.71	16.06	23.77	12.51	12.18	9.96
<u>Salary</u>									
in £1,000	29.34	29.57	29.31	29.45	29.53	31.14	28.81	28.74	28.22
Deviation									
Percent	0.00	0.81	-0.10	0.39	0.64	6.16	-1.81	-2.05	-3.82

Table B.2: simulations on individual and productivity characteristics (2)

Appendix C: Comparable wage regressions

In this appendix we calculate straightforward wage regression to compare them to the results of our spinal point and salary scale model. Define w_i as the natural logarithm of the salary of individual i, x_i as a vector of explanatory variables, β as a parameter vector, and ε_i^w as an individual disturbance term.

(B.1)
$$w_i = x_i^{\beta} + \varepsilon_i^{W}$$

The OLS results are given in table C.1. We consider salary determination of our full sample of academics, of academics excluding professors and of professors only. Note that the results of the last two regressions should be interpreted with care, as selection effects play a role.

For the full sample we reveal an insignificant reward to male academics above female. As for the salary scale model, excluding the productivity variables from the analysis reveals a significant gender-wage gap at a one percent significance level. Experience is positively rewarded and spells outside the labour market have a significantly negative effect on academic salaries. Academics in Heriot-Watt experience a significant salary advantage relative to the excluded university Glasgow. Results reveal significantly positive rewards to productivity variables such as the number of books and papers published, grants awarded and high teaching ability. Overall the conclusions are in line with the results from the salary scale model.

Comparison of these results with those of academics excluding professors reveals similar patterns, although the reward to tenure is now significant. As stated in the beginning of this paragraph, the results should be taken with care as selection effects might play a major role here. For academics with much experience, tenure, and publications, becoming professor is a likely event. As the professors are excluded, the impact of these variables might be biased considerably. The same holds for the regression on the wages of the professors. Notice the negative impact of the number of books written, and also the n-shaped effect of tenure is negative after 8 years. Although this result is in line with Ransom (1993), it does not seem very reasonable to draw strong conclusions on the basis of these results.

	Full s	sample	Exclud	l. Professors	Profes	sors only	
	(635	obs.)	(544 c	(544 obs.)		(91 obs.)	
	par.	s.e	par.	s.e.	par.	s.e.	
Individual char.							
Intercept	9.5323	***(0.0270)	9.5565	***(0.0236)	10.4829	***(0.1708)	
Gender	0.0243	(0.0149)	0.0010	(0.0127)	-0.0637	(0.0827)	
Citizen	0.0263	(0.0190)	0.0094	(0.0161)	-0.0229	(0.0643)	
Job char.							
Experience	0.0329	***(0.0025)	0.0267	***(0.0023)	0.0057	(0.0093)	
Experience ² /10	-0.0045	***(0.0006)	-0.0039	***(0.0006)	0.0001	(0.0016)	
Tenure	0.0029	(0.0027)	0.0122	***(0.0026)	0.0043	(0.0042)	
Tenure ² /10	-0.0011	(0.0009)	-0.0028	***(0.0008)	-0.0027	**(0.0013)	
Time-out	-0.0171	***(0.0037)	-0.0116	***(0.0031)	-0.0642	(0.0425)	
<u>University</u>							
Aberdeen	0.0254	(0.0185)	0.0219	(0.0172)	0.0338	(0.0335)	
Dundee	-0.0050	(0.0162)	0.0110	(0.0142)	-0.0047	(0.0422)	
Heriot-Watts	0.0599	**(0.0262)	0.0685	***(0.0250)	-0.0054	(0.0408)	
St.Andrews	-0.0008	(0.0180)	-0.0041	(0.0162)	0.0716	**(0.0334)	
Faculty							
Arts	0.0089	(0.0184)	0.0111	(0.0166)	-0.0287	(0.0355)	
Engineer	-0.0302	(0.0199)	-0.0379	**(0.0178)	0.0328	(0.0422)	
Medicine	0.0090	(0.0199)	0.0020	(0.0177)	0.0759	*(0.0447)	
Soc.Science	0.0500	***(0.0191)	0.0428	**(0.0176)	-0.0122	(0.0348)	
Publications							
Books / average by fac.	0.0096	***(0.0032)	0.0060	(0.0040)	-0.0061	*(0.0034)	
Chapt. / average by fac.	-0.0013	(0.0025)	0.0005	(0.0025)	-0.0036	(0.0040)	
Papers / average by fac.	0.0443	***(0.0058)	0.0325	***(0.0074)	0.0025	(0.0064)	
<u>Other</u>							
Grants / average by fac.	0.0231	***(0.0045)	0.0165	***(0.0045)	0.0103	(0.0068)	
Having PhD	0.0246	*(0.0146)	0.0449	***(0.0132)	-0.0494	(0.0311)	
Teach.skill	0.0566	***(0.0157)	0.0507	***(0.0147)	0.0480	*(0.0268)	

Table C.1: Estimation results of wage regressions

IZA Discussion Papers

No.	Author(s)	Title	Area	Date
81	T. J. Hatton S. Wheatley Price	Migration, Migrants and Policy in the United Kingdom	1	12/99
82	K. A. Konrad	Privacy, time consistent optimal labor income taxation and education policy	3	12/99
83	R. Euwals	Female Labour Supply, Flexibility of Working Hours, and Job Mobility in the Netherlands	1	12/99
84	C. M. Schmidt	The Heterogeneity and Cyclical Sensitivity of Unemployment: An Exploration of German Labor Market Flows	1	12/99
85	S. Pudney M. A. Shields	Gender and Racial Discrimination in Pay and Promotion for NHS Nurses	5/6	12/99
86	J.P. Haisken-DeNew C. M. Schmidt	Money for Nothing and Your Chips for Free? The Anatomy of the PC Wage Differential	5	12/99
87	T. K. Bauer	Educational Mismatch and Wages in Germany	1	12/99
88	O. Bover P. Velilla	Migration in Spain: Historical Background and Current Trends	1	12/99
89	S. Neuman	Aliyah to Israel: Immigration under Conditions of Adversity	1	12/99
90	H. Lehmann J. Wadsworth	Tenures that Shook the World: Worker Turnover in Russia, Poland and Britain	4	12/99
91	M. Lechner	Identification and Estimation of Causal Effects of Multiple Treatments Under the Conditional Independence Assumption	6	12/99
92	R. E. Wright	The Rate of Return to Private Schooling	5	12/99
93	M. Lechner	An Evaluation of Public-Sector-Sponsored Continuous Vocational Training Programs in East Germany	6	12/99
94	M. Eichler M. Lechner	An Evaluation of Public Employment Programmes in the East German State of Sachsen-Anhalt	6	12/99
95	P. Cahuc A. Zylberberg	Job Protection, Minimum Wage and Unemployment	3	12/99
96	P. Cahuc A. Zylberberg	Redundancy Payments, Incomplete Labor Contracts, Unemployment and Welfare	3	12/99

97	A. Barrett	Irish Migration: Characteristics, Causes and Consequences	1	12/99
98	J.P. Haisken-DeNew C. M. Schmidt	Industry Wage Differentials Revisited: A Longitudinal Comparison of Germany and USA	1/5	12/99
99	R. T. Riphahn	Residential Location and Youth Unemployment: The Economic Geography of School-to-Work- Transitions	1	12/99
100	J. Hansen M. Lofstrom	Immigrant Assimilation and Welfare Participation: Do Immigrants Assimilate Into or Out-of Welfare?	1/3	12/99
101	L. Husted H. S. Nielsen M. Rosholm N. Smith	Employment and Wage Assimilation of Male First Generation Immigrants in Denmark	3	1/00
102	B. van der Klaauw J. C. van Ours	Labor Supply and Matching Rates for Welfare Recipients: An Analysis Using Neighborhood Characteristics	2/3	1/00
103	K. Brännäs	Estimation in a Duration Model for Evaluating Educational Programs	6	1/00
104	S. Kohns	Different Skill Levels and Firing Costs in a Matching Model with Uncertainty – An Extension of Mortensen and Pissarides (1994)	1	1/00
105	G. Brunello C. Graziano B. Parigi	Ownership or Performance: What Determines Board of Directors' Turnover in Italy?	1	1/00
106	L. Bellmann S. Bender U. Hornsteiner	Job Tenure of Two Cohorts of Young German Men 1979 - 1990: An analysis of the (West-)German Employment Statistic Register Sample concerning multivariate failure times and unobserved heterogeneity	1	1/00
107	J. C. van Ours G. Ridder	Fast Track or Failure: A Study of the Completion Rates of Graduate Students in Economics	5	1/00
108	J. Boone J. C. van Ours	Modeling Financial Incentives to Get Unemployed Back to Work	3/6	1/00
109	G. J. van den Berg B. van der Klaauw	Combining Micro and Macro Unemployment Duration Data	3	1/00
110	D. DeVoretz C. Werner	A Theory of Social Forces and Immigrant Second Language Acquisition	1	2/00
111	V. Sorm K. Terrell	Sectoral Restructuring and Labor Mobility: A Comparative Look at the Czech Republic	1/4	2/00
112	L. Bellmann T. Schank	Innovations, Wages and Demand for Heterogeneous Labour: New Evidence from a Matched Employer-Employee Data-Set	5	2/00
113	R. Euwals	Do Mandatory Pensions Decrease Household Savings? Evidence for the Netherlands	3	2/00

114	G. Brunello A. Medio	An Explanation of International Differences in Education and Workplace Training	2	2/00
115	A. Cigno F. C. Rosati	Why do Indian Children Work, and is it Bad for Them?	3	2/00
116	C. Belzil	Unemployment Insurance and Subsequent Job Duration: Job Matching vs. Unobserved Heterogeneity	3	2/00
117	S. Bender A. Haas C. Klose	IAB Employment Subsample 1975-1995. Opportunities for Analysis Provided by the Anonymised Subsample	7	2/00
118	M. A. Shields M. E. Ward	Improving Nurse Retention in the British National Health Service: The Impact of Job Satisfaction on Intentions to Quit	5	2/00
119	A. Lindbeck D. J. Snower	The Division of Labor and the Market for Organizations	5	2/00
120	P. T. Pereira P. S. Martins	Does Education Reduce Wage Inequality? Quantile Regressions Evidence from Fifteen European Countries	5	2/00
121	J. C. van Ours	Do Active Labor Market Policies Help Unemployed Workers to Find and Keep Regular Jobs?	4/6	3/00
122	D. Munich J. Svejnar K. Terrell	Returns to Human Capital under the Communist Wage Grid and During the Transition to a Market Economy	4	3/00
123	J. Hunt	Why Do People Still Live in East Germany?	1	3/00
124	R. T. Riphahn	Rational Poverty or Poor Rationality? The Take-up of Social Assistance Benefits	3	3/00
125	F. Büchel J. R. Frick	The Income Portfolio of Immigrants in Germany - Effects of Ethnic Origin and Assimilation. Or: Who Gains from Income Re-Distribution?	1/3	3/00
126	J. Fersterer R. Winter-Ebmer	Smoking, Discount Rates, and Returns to Education	5	3/00
127	M. Karanassou D. J. Snower	Characteristics of Unemployment Dynamics: The Chain Reaction Approach	3	3/00
128	O. Ashenfelter D. Ashmore O. Deschênes	Do Unemployment Insurance Recipients Actively Seek Work? Evidence From Randomized Trials in Four U.S. States	6	3/00
129	B. R. Chiswick M. E. Hurst	The Employment, Unemployment and Unemployment Compensation Benefits of Immigrants	1/3	3/00
130	G. Brunello S. Comi C. Lucifora	The Returns to Education in Italy: A New Look at the Evidence	5	3/00
131	B. R. Chiswick	Are Immigrants Favorably Self-Selected? An Economic Analysis	1	3/00

132	R. A. Hart	Hours and Wages in the Depression: British Engineering, 1926-1938	7	3/00
133	D. N. F. Bell R. A. Hart O. Hübler W. Schwerdt	Paid and Unpaid Overtime Working in Germany and the UK	1	3/00
134	A. D. Kugler G. Saint-Paul	Hiring and Firing Costs, Adverse Selection and Long-term Unemployment	3	3/00
135	A. Barrett P. J. O'Connell	Is There a Wage Premium for Returning Irish Migrants?	1	3/00
136	M. Bräuninger M. Pannenberg	Unemployment and Productivity Growth: An Empirical Analysis within the Augmented Solow Model	3	3/00
137	JSt. Pischke	Continuous Training in Germany	5	3/00
138	J. Zweimüller R. Winter-Ebmer	Firm-specific Training: Consequences for Job Mobility	1	3/00
139	R. A. Hart Y. Ma	Wages, Hours and Human Capital over the Life Cycle	1	3/00
140	G. Brunello S. Comi	Education and Earnings Growth: Evidence from 11 European Countries	2/5	4/00
141	R. Hujer M. Wellner	The Effects of Public Sector Sponsored Training on Individual Employment Performance in East Germany	6	4/00
142	J. J. Dolado F. Felgueroso J. F. Jimeno	Explaining Youth Labor Market Problems in Spain: Crowding-Out, Institutions, or Technology Shifts?	3	4/00
143	P. J. Luke M. E. Schaffer	Wage Determination in Russia: An Econometric Investigation	4	4/00
144	G. Saint-Paul	Flexibility vs. Rigidity: Does Spain have the worst of both Worlds?	1	4/00
145	MS. Yun	Decomposition Analysis for a Binary Choice Model	7	4/00
146	T. K. Bauer J. P. Haisken-DeNew	Employer Learning and the Returns to Schooling	5	4/00
147	M. Belot J. C. van Ours	Does the Recent Success of Some OECD Countries in Lowering their Unemployment Rates Lie in the Clever Design of their Labour Market Reforms?	3	4/00
148	L. Goerke	Employment Effects of Labour Taxation in an Efficiency Wage Model with Alternative Budget Constraints and Time Horizons	3	5/00
149	R. Lalive J. C. van Ours J. Zweimüller	The Impact of Active Labor Market Programs and Benefit Entitlement Rules on the Duration of Unemployment	3/6	5/00

150	J. DiNardo K. F. Hallock JSt. Pischke	Unions and the Labor Market for Managers	7	5/00
151	M. Ward	Gender, Salary and Promotion in the Academic Profession	5	5/00
152	J. J. Dolado F. Felgueroso J. F. Jimeno	The Role of the Minimum Wage in the Welfare State: An Appraisal	3	5/00
153	A. S. Kalwij M. Gregory	Overtime Hours in Great Britain over the Period 1975-1999: A Panel Data Analysis	3	5/00
154	M. Gerfin M. Lechner	Microeconometric Evaluation of the Active Labour Market Policy in Switzerland	6	5/00
155	J. Hansen	The Duration of Immigrants' Unemployment Spells: Evidence from Sweden	1/3	5/00
156	C. Dustmann F. Fabbri	Language Proficiency and Labour Market Per- formance of Immigrants in the UK	1	5/00
157	P. Apps R. Rees	Household Production, Full Consumption and the Costs of Children	7	5/00
158	A. Björklund T. Eriksson M. Jäntti O. Raaum E. Österbacka	Brother Correlations in Earnings in Denmark, Finland, Norway and Sweden Compared to the United States	5	5/00
159	P J. Jost M. Kräkel	Preemptive Behavior in Sequential Tournaments	5	5/00
160	M. Lofstrom	A Comparison of the Human Capital and Signaling Models: The Case of the Self-Employed and the Increase in the Schooling Premium in the 1980's	5	6/00
161	V. Gimpelson D. Treisman G. Monusova	Public Employment and Redistributive Politics: Evidence from Russia's Regions	4	6/00
162	C. Dustmann M. E. Rochina- Barrachina	Selection Correction in Panel Data Models: An Application to Labour Supply and Wages	6	6/00
163	R. A. Hart Y. Ma	Why do Firms Pay an Overtime Premium?	5	6/00
164	M. A. Shields S. Wheatley Price	Racial Harassment, Job Satisfaction and Intentions to Quit: Evidence from the British Nursing Profession	5	6/00
165	P. J. Pedersen	Immigration in a High Unemployment Economy: The Recent Danish Experience	1	6/00
166	Z. MacDonald M. A. Shields	The Impact of Alcohol Consumption on Occupa- tional Attainment in England	5	6/00

167	A. Barrett J. FitzGerald B. Nolan	Earnings Inequality, Returns to Education and Immigration into Ireland	5	6/00
168	G. S. Epstein A. L. Hillman	Social Harmony at the Boundaries of the Welfare State: Immigrants and Social Transfers	3	6/00
169	R. Winkelmann	Immigration Policies and their Impact: The Case of New Zealand and Australia	1	7/00
170	T. K. Bauer K. F. Zimmermann	Immigration Policy in Integrated National Economies	1	7/00
171	C. Dustmann F. Windmeijer	Wages and the Demand for Health – A Life Cycle Analysis	5	7/00
172	D. Card	Reforming the Financial Incentives of the Welfare System	3	7/00
173	D. S. Hamermesh	Timing, Togetherness and Time Windfalls	5	7/00
174	E. Fehr JR. Tyran	Does Money Illusion Matter? An Experimental Approach	7	7/00
175	M. Lofstrom	Self-Employment and Earnings among High- Skilled Immigrants in the United States	1	7/00
176	O. Hübler W. Meyer	Industrial Relations and the Wage Differentials between Skilled and Unskilled Blue-Collar Workers within Establishments: An Empirical Analysis with Data of Manufacturing Firms	5	7/00
177	B. R. Chiswick G. Repetto	Immigrant Adjustment in Israel: Literacy and Fluency in Hebrew and Earnings	1	7/00
178	R. Euwals M. Ward	The Renumeration of British Academics	5	7/00